Author’s response to reviews

Title: Assessment of the knowledge of public and private physicians regarding Tuberculosis management in Karachi

Authors:

Maliha Naseer (maliha.naseer@aku.edu)
Ali Khawaja (alianwar90@hotmail.com)
Amin Pethani (amin.pethani@aku.edu)
Salik Aleem (salikaleem@hotmail.com)

Version: 3 Date: 29 June 2013

Author’s response to reviews: see over
Dear Editor,

We are extremely grateful to the reviewers for their constructive feedback on our manuscript. All the changes made in the manuscript are highlighted RED.

Point by point reply to reviewers’ comments is as follows:

**Reviewer 1:**

1. Please include a logistic regression after Table 4. Also change the title of Table 4 to reflect the data presented i.e. remove logistic regression analysis since it was not done in this Table. I would rather suggest the title: Factors affecting the knowledge of physicians regarding tuberculosis management in urban Pakistan.

   **Ans:** Title of table 4 has been changed as suggested by reviewers and table 5 made to show logistic regression analysis with regression coefficient.

2. Fig 1 appeared twice. Remove one of them.

   **Ans:** This has been done as suggested by the reviewer.

3. Include your study limitations as a subsection under the method section.

   **Ans:** We have included the limitations at the end of the discussion section.

4. Please indicate in the Tables which p values are significant (i.e. where p value is less than 0.05).

   **Ans:** This has been done as suggested by the reviewer.

**Reviewer 2:**

No further revisions were suggested. We thank the reviewer for his constructive feedback.

With best regards,

Maliha Naseer
Email: maliha.naseer@gmail.com.